Author’s response to reviews

Title: Difference between pre-operative and cardiopulmonary bypass mean arterial pressure is independently associated with early cardiac surgery-associated acute kidney injury

Authors:

Hussein D Kanji (hdkanji@gmail.com)
Costas J Schulze (costas.schulze@gmail.com)
Marilou Hervas-Malo (marilou.hervas@ualberta.ca)
Peter Wang (peter.wang@ualberta.ca)
David B Ross (David.B.Ross@albertahealthservices.ca)
Mohamad Zibdawi (mohamad.zibdawi@albertahealthservices.ca)
Sean M Bagshaw (bagshaw@ualberta.ca)

Version: 3 Date: 24 August 2010

Author’s response to reviews:

Dr. V. Zamvar
Editor-In-Chief
Journal of Cardiothoracic Surgery
August 17, 2010

Dear Dr. Zamvar:

RE: Manuscript 8518520443895232 submission titled “Difference Between Pre-operative and Cardiopulmonary Bypass Mean Arterial Pressures is Independently Associated with Early Cardiac Surgery-Associated Acute Kidney Injury”

We have completed the requested format changes.

If there is any further information required, please contact us at your convenience.

Sincerely,

Sean M Bagshaw, MD, MSc